DOI: 10.1002/aur.2583

RESEARCH ARTICLE

Use of allied-health services and medication among adults with ASD in Latin America

Andy Torres¹ | Denise Lecusay¹ | Daniel Valdez² | Analía Rosoli³ | Gabriela Garrido⁴ | Sebastián Cukier⁵ | Cristiane Silvestre Paula⁶ | Ricardo Garcia⁷ | Alexia Rattazzi⁵ | Cecilia Montiel-Nava¹

¹Department of Psychological Sciences, University of Texas Rio Grande Valley, Edinburg, Texas, USA

²Universidad de Buenos Aires, Buenos Aires, Argentina

³Organización Estados Iberoamericanos para la Educación la Ciencia y la Cultura (OEI), Santo Domingo, Dominican Republic

⁴Universidad de la República, Montevideo, Uruguay

⁵Programa Argentino para Niños, Adolescentes y Adultos con Condiciones del Espectro Autista, Buenos Aires, Argentina

⁶Programa de Pós-Graduação em Distúrbios do Desenvolvimento Universidade Presbiteriana Mackenzie (UPM), São Paulo, SP, Brazil

⁷Universidad de Chile, Santiago, Chile

Universidad de Chile, Santiago, Chile

Correspondence

Cecilia Montiel-Nava, Department of Psychological Sciences, University of Texas Rio Grande Valley, 1200 University Drive, Edinburg, TX 78541, USA. Email: cecilia.montielnava@utrgv.edu

Abstract

ASD is a neurodevelopmental disorder that affects people across the entire lifespan, yet most of the research identifying the health and medical needs for autistic individuals have been among minors. As individuals with ASD transition to emerging adulthood, the services, already limited, become less available. This study aimed to identify the use of services for Latin American adults on the spectrum. We surveyed 295 caregivers of adults with ASD from six Latin American countries. Comparing the results for all the possible services observed in this study, the adults in our sample were primarily underserved: 84.4%-95.9% were receiving zero hours per week, 3.7%-12.9% 1%-10 h, 0%-1.7% 11–20, and only 0%-1% above 20 h of services. Almost half of the sample used medication, and neurologists were the most consulted health providers. Next to inexistent health care usage in Latin American adults with ASD highlights socioeconomic and health disparities in service provision for ASD in the region. The lack of services places adults with ASD in Latin America at a higher risk of worse symptom severity than autistic adults from regions with broader access to services.

Lay Summary

This study aimed to identify the quantity of services received by adults with autism in Latin America. Most of our sample was not receiving health services yet almost half had access to medication. This could mean that adults with autism in Latin America are at higher risk for poorer health.

KEYWORDS

allied health, autism, autism medication, health disparities, Hispanics, Latin America

INTRODUCTION

Autism spectrum disorder (ASD) is a lifelong neurodevelopmental disorder primarily characterized by socio-communication deficits and the presence of restricted, repetitive patterns of behavior and interests (American Psychiatric Association, 2013). Adults with ASD are not only at a higher risk for poorer health outcomes and medical complications, but also have a shorter life expectancy compared to the general population (Hirvikoski et al., 2016; Jokiranta-Olkoniemi et al., 2020; Mandell, 2018). ASD is thus a disability that requires

© 2021 International Society for Autism Research and Wiley Periodicals LLC.

external support throughout adulthood for many of those affected by the condition (Roux et al., 2015; Croen et al., 2015; Nicolaidis et al., 2014; Murphy et al., 2016, Brugha et al., 2011). As individuals with ASD transition to emerging adulthood, the services, already limited, become less available (Płatos & Pisula, 2019; Shattuck et al., 2012).

Symptom expression in both verbal and non-verbal individuals with ASD is displayed heterogeneously throughout the population, making it imperative that service provision for adults with ASD be tailored to their individual needs (Mason et al., 2019). Barriers to

.....

accessing healthcare services for adults might vary from a combination of factors such as policy restrictions set forth by insurance companies, limited availability of services, and lack of ASD knowledge by providers and the public (Raymaker et al., 2017, Shattuck et al., 2020; Day et al., 2020; Lake, Perry & Lunsky, 2014; Day et al., 2020; Friedman et al., 2013). Navigating medical care is more cumbersome for autistic adults given the reported limited awareness of various healthcare system aspects such as insurance coverage, billing, or medical benefits (Brugha et al., 2011; Shattuck et al., 2012). Such barriers and challenges could be partially influenced by limited knowledge of transitioning services from families and caregivers of autistic adults as they might not be able to advocate for such services or help their relatives navigate the healthcare system (Cheak-Zamora & Thullen, 2017; Croen et al., 2015; Raymaker et al., 2017). Incorrect assumptions about needs or skills adults with ASD and inappropriate language use by the provider toward the ASD adult can decrease future patient autonomy and treatment seeking behavior (Kinnear et al., 2016; Obeid et al., 2015).

Autism symptoms, mental and physical health issues, and behavioral challenges are often left untreated in adulthood due to insufficient medical coverage or lack thereof after transitioning (Hansel, 2013; Friedman et al., 2013; Richards et al., 2012). Moreover, those autistic adults with medical problems may experience more barriers to service use than those with no medical problems (Vogan et al., 2017). Among the barriers, the high coverage costs, both out-of-pocket and deductible fees, remain temporally constant during the developmental trajectory of individuals with ASD (Buck et al., 2014; Kennedy-Hendricks et al., 2018; LeClerc & Easley, 2015).

On the other hand, medication use in individuals with ASD also increases with age (Buck et al., 2014; Fusar-Poli et al., 2019). The most-reported medications include neuroleptics, anticonvulsants, and antidepressants (Buck et al., 2014; Daniels et al., 2008). Because of adults' various health conditions with autism, medication for those co-occurring conditions can be helpful. However, delays in diagnosing and problems recognizing symptoms might cause some of these conditions to be over or undertreated. (Espadas et al., 2020; Mandell et al., 2008).

The limited research on autism in low-and-middleincome countries complicates the understanding of whether individuals with ASD do not receive services because what they need is not available or because they require different or more services (Montiel-Nava et al., 2020). Uneven distribution of social, economic, political, and environmental resources, like those experienced in Latin American countries, are risk factors for health disparities (Rubin et al., 2011). This paper presents data on six different Latin American countries considered as upper-middle or high-income countries according to the World Bank classification (World Bank, 2017). However, people in these countries hardly ever have access to services and trained specialists since there is a shortage of mental health professionals, and government mental health expenditure is below standard (World Health Organization, WHO, 2017). Most of the participating countries reports universal health coverage for mental health services (WHO, 2005, 2017). However, this universal health coverage implies mainly overbooked urban facilities, few trained professionals, and long waiting lists (Kohn et al., 2005; Saldivia et al., 2004).

Systematic data about ASD in adulthood is limited for Latin American countries (Baxter et al., 2015; Paula et al., 2020) and U.S. populations. For instance, most of the literature on ASD in Latin America (Paula et al., 2011; (Montiel-Nava & Pena, 2008) (Bagaiolo et al., 2017; Bordini et al., 2020) and for Hispanics living in the U.S. focuses on children. (Zuckerman et al., 2014). Although the symptom severity is similar regardless of race or ethnicity (Magaña & Smith, 2006), other factors like psychological well-being and/or adaptive function could be considered a disadvantage. Latin American families with an autistic member exhibit health inequality in the form of long periods of wait time to access services, inability to pay for the costs of services, and overall limited availability of ASD services (Paula et al., 2020). Therefore, existing socioeconomic disparities affect the initial identification and diagnostic process of ASD in Latinos (Montiel-Nava et al., 2017) but also prevents the access or continuation of ASD services, which can have adverse physical health outcomes for the adult with ASD (Shattuck et al., 2020). The overall health of Latin American adults with ASD is likely to be significantly poorer than autistics from developing countries, as is the limited ASD-services encounters among Latin American children and adolescents (Montiel-Nava et al., 2020).

In summary, adults with ASD living in Latin American countries are at a higher risk of being under-diagnosed, underserved, and experience a lower quality of services than adults with ASD living in developed countries. This study aimed to identify the usage of health services and medication by adults with ASD in Latin America.

METHODS

This study was carried out in six different countries in South and Central America: Argentina, Brazil, Chile, Dominican Republic, Uruguay, and Venezuela. Advertisement of the study was disseminated through social networking platforms, parent and professional associations, institutions that provide services for individuals with ASD. A sample of convenience was drawn from each country. This study is part of a multisite study of Caregivers Needs in Latin America sponsored by the REAL network (Red Espectro Autista Latinoamerica, Montiel-Nava et al., 2020). Due to the disparity in ages, data on caregivers of adult children were analyzed separately.

Participants

This study, an online anonymous survey, was carried out in six different countries in South and Central America: Argentina, Brazil, Chile, Dominican Republic, Uruguay, and Venezuela. Advertisement of the study was disseminated through social networking platforms, parent and professional associations, institutions that provide services for individuals with ASD. A sample of convenience was drawn from each country. This study is part of a multisite study of Caregivers Needs in Latin America sponsored by the REAL network (Red Espectro Autista Latinoamerica, Montiel-Nava et al., 2020). Due to the disparity in ages, data on caregivers of adult children were analyzed separately.

Instruments

Caregivers completed the questionnaire "Caregiver Needs Survey" (Daniels et al., 2017). This survey was designed to be used by parents or primary caregivers of individuals diagnosed with ASD. The survey contained four sections: family demographics, affected individual characteristics, service encounters, and parent/caregiver perceptions. Questions for the section on service encounters included past and current service use for a variety of ASD-specific and non-ASD-specific services as well as education and educational supports (adapted from the Survey of Pathways to Diagnosis and Services; CDC, 2011). For this study, REAL translated the questionnaire into both Spanish and Portuguese; and translated it back to the original language (English) to assure that the instrument's integrity was preserved through translation. After the backtranslation was completed, each country made modifications to items related to educational modalities and levels, insurance coverage types, and informed consent details to ensure validity for cultural contexts. In general, language adjustments were responsible for most changes and adaptations to the survey.

Caregivers entered the website and were asked to choose the country of residence from a cascading menu. The selection opened a separate window with informed consent for their country, and initials were requested as a signature. Once participants signed the informed consent, they received a random computer-generated ID, which offered caregivers the opportunity to come back any time to complete the survey without losing the entered data. No personal information was required (i.e., name, phone number, email address, etc.), preserving the anonymity of all participants and confidentiality of the data. Ethics approval was granted by the ethics board of the different agencies to which the national coordinator was affiliated.

The caregiver completed all items about the person with ASD. The sociodemographic measures included sex of the individual with ASD, insurance coverage (private, public, or both), and country. Clinical variables referred to the diagnosis of the affected individual (autism/autistic disorder, Asperger's disorder, PDD, PDD-NOS, ASD, and other), and the presence of medical comorbidities at the time of the study. The survey contains diagnosis options based on the DSM-IV-TR (APA, 2000) and the DSM-5 (APA, 2013).

Participating caregivers indicated (a) current service encounters from a list of allied- health services for their children and (b) the number of hours per week for each current service for their children at the time of the study. The original survey included eight allied- health services, Speech Therapy, Occupational Therapy, Physical Therapy, Behavioral Therapy, Cognitive Rehabilitation, Sensory Integration, Social Skills Training, Psychoanalytic Therapy, and medication frequency. The original survey also contains Relational Therapies (e.g., Floortime, SUNRISE) but was excluded from the analysis due to low numbers (n = 1).

The survey used three different questions to inquire about the autistic adults' language, adaptive function, and intellectual level. The questions were, (1) What is the verbal ability of your child? With response options including "(a) Does not speak, (b) uses single words, (c) uses 2-3-word phrases, (d) uses sentences of four or more words, and (e) uses full sentences." (2) In your opinion, how is the level of functionality and severity of your child? The response options were, (a) mild symptomatology/high function), (b) moderate symptomatology/moderate function, (c) severe symptomatology/low function, and (d) I do not know. The final question, (3) Based on the professionals' information, what is the intellectual level (IQ/intelligence) of your child? with response options including "(a) severe, (b) mild, (c) average, (d) above average, and (e) I do not know." The caregivers responded to the questionnaire about individuals ranging from nonverbal to verbally fluent. However, different individuals' verbal and intellectual development and functioning level scores may not be directly comparable. For that purpose, the three variables (language, severity, and adaptive function) were transformed to severity scores combining the three variables to achieve comparability across the subjects.

Data analysis

Data sets were reviewed for duplicate or missing cases, which were excluded from the analysis. Duplicates were verified by comparing initials used in the consent form, country and region of residence indicated in the survey, and the caregivers' IP addresses (Montiel-Nava et al., 2020; Paula et al., 2020).

Data analysis included calculations of central tendency and frequency. Country, sex of the individual with ASD (male or female), another family member with ASD, insurance type, current diagnosis, medical comorbidities, and type of service were treated as nominal; severity index and usage of services were treated as ordinal. Age and age of diagnosis were treated as continuous variables. Group comparisons (based on age groups) were conducted using chi-square tests for categorical variables and independent sample T-tests for continuous variables.

Usage of services, our main dependent variable, was measured as the current (at the time of participation) number of hours received of health service (or the current medication frequency).

Age group

As emerging adults transition to higher education from secondary education (Elias & White, 2018; Liptak et al., 2011) and into the workforce (Griffiths et al., 2016; Rogers & Vismara, 2008), they are faced with new challenges related to their independence (Friedman et al., 2013). In contrast, middle-aged adults face a new set of challenges associated with the aging process (Stuart-Hamilton & Morgan, 2011). Those changes might include neuroanatomy (Courchesne et al., 2011; Zielinski et al., 2014), neurocognitive function (Braden et al., 2017), and worsening health and well-being. Therefore, we categorize autistic adults by age groups into emerging adults (18–25 years) and middle-aged adults (26–50 years).

Symptom severity

A four-step procedure was used to derive the severity score: (a) normalization of data using z-scores; (b) addition of the z-scores [(score-mean)/SD]; (c) defining the severity range: +3 SD p 99.9 (severe), +2 SD p 98 (moderate), and +1 SD p 84 (low); and (d) performance of z-score transformation up to three ranks: low/medium/high severity. The index is directly proportional, that is, a higher severity index indicates a higher severity in symptoms.

RESULTS

Sample characteristics

The total sample was 295 caregivers from the different participant countries, and most of the caregivers were from Brazil 56.3% (n = 166). The mean age of their children (adults) was 24.3 years (*SD* 6.6 years), and ages ranged from 18 to 50 years. In terms of adulthood stage, 68.13% (n = 201) were emerging adults (18–25 years old), and 31.87% (n = 94) were middle-aged adults (26–50 years old). Caregivers were primarily mothers (73.9%), with undergraduate (55.5%) or graduate (33.6%) degrees. While most of the caregivers were mothers (73.9%, n = 218), most of their adult children were identified as males 75.6% (n = 223) (Table 1). From the total

sample of caregivers, 95.9% (n = 283) had health insurance, 68.5% (n = 202) had only private, 22.7% (n = 67) had only public, 4.7% (n = 14) had both private and public, and 4.1% (n = 12) did not respond the item. There was no relationship between type of coverage and age groups ($x^2 = 0.94$, p > 0.05).

Clinical characteristics

The most prevalent diagnosis of the adult child was Asperger Disorder (34.90%, n = 103), followed by Autism/Autistic Disorder (28.8%, n = 85). The least common diagnosis was PDD (4.1%, n = 12). Medical comorbidities analysis shows epilepsy is the most frequent condition among the sample (18.6%, n = 55). Most adults (73.2%, n = 201) had at least one comorbid diagnosis (besides the autism-related diagnosis). The comorbid disorders included Epilepsy (18.6%, n = 55), Down Syndrome (0.7%, n = 2), Fragile X (2.4%, n = 7), and others (22.4%, n = 66). For the "Other" category, the survey had a blank space for caregiver to fill. Responses included depression, ADHD, genetic and congenital disorders, and respiratory/allergy-related conditions. The sample was equally distributed between Moderate (40%, n = 118) and High (40%, n = 118) severity. None of the clinical variables reached significant differences across age groups. (Table 2).

Allied-health services and medication

Behavior therapy was the most frequently used alliedhealth service; however, 84.4% (n = 249) were not receiving this service. For those receiving behavior therapy, only 2.7% (n = 8) received 11+ hours per week. The least frequent service was Sensory Integration Therapy, where 95.9% (n = 283) did not receive this service, and for those that did receive it, none were in the 11+11+ hours per week category (Table 3). Only one of the services showed a significant difference among the two age groups; emerging adults were more likely than middle-aged adults to receive Social Skills Training ($X^2 = 7.1, p < 0.05$). On the other hand, 48.6% (n = 143) of participants were using medication at the time of the study, with no significant differences across age groups (Table 3). Regarding the overall number of utilized health services, 58% (n = 171) of participants were receiving zero types of services,), while 16.9% (n = 50) were receiving at least one type of allied-health service. There were no significant differences in the number allied-health services utilized by the two age groups (Table 3).

The chi-square analysis of the relationship between medication and medical comorbidities showed that most of those receiving medication had only one comorbid diagnosis (57.8%, n = 170) (Table 4). Finally, the most utilized health providers were Neurologists

TABLE 1 Distribution of demographic characteristics of the sample

	Emerging adults (Age: 18–25) $n = 201$	Middle-aged adults (Age: 26–50) $n = 94$	Total <i>n</i> = 295
Who completed the survey			
Mother	79.5 (158)	63.8 (60)	218 (73.9)
Father	11 (22)	10.6 (10)	32 (10.9)
Grandparents	0.5 (1)	1.1 (1)	2 (0.7)
Other	9 (18)	24.5 (23)	41 (14.5)
Highest educational level			
Elementary or less	11.5 (23)	9.7 (9)	10.9 (32)
High School Diploma	19 (38)	12.9 (12)	17.1 (50)
Undergraduate Degree	36.5 (73)	43 (40)	38.6 (113)
Graduate Degree	33 (66)	34.4 (32)	33.4 (98)
Country			
Argentina	21.4 (43)	11.7 (11)	18.3 (54)
Brazil	52.2 (105)	64.9 (61)	56.3 (166)
Chile	10 (20)	7.4 (7.4)	9.2 (27)
Dominican Republic	2.5 (5)	0 (0)	1.7 (5)
Uruguay	11.4 (23)	12.8 (12)	11.9 (35)
Venezuela	2.5 (5)	0 (0)	1.7 (5)
Type of insurance coverage			
Only private	70.8 (136)	72.5 (66)	71.4
Only public	24 (46)	23.1 (21)	23.7 (67)
Both private and public	5.2 (10)	4.4 (4)	4.9 (14)
Family member with an ASD diagnosis n (%)	77.6 (156)	89.4 (84)	90.8 (268)
Adults with ASD demographics gender [N (%)]			
Male	96.5 (194)	94.7 (89)	95.9 (283)
Female	3 (6)	5.3 (5)	3.7 (11)
Age, Mean \pm SD	20.58 ± 2	31.1 ± 6.2	24.3 ± 6.6

(44.3%, n = 129), followed by dietitians (35.7%, n = 104). The least used professionals were psychiatrists and (14.6, n = 42), and audiologists (6.3%, n = 18). There were no significant differences by age group in any of our identified medical and health service providers. (Table 5).

Supplemental analysis to look for possible predictors of use of service were all non-significant (Tables S1 and S2). There were no significant differences in allied-health usage by either country, gender, caregivers' educational level, diagnosis, or insurance coverage type. While social skills training revealed to have a significant difference by countries in the Chi-Square analysis, post-hoc testing revealed non-significant differences, as the standardized residuals all fail to reach statistical significance (adj. Standardize values <1.96).

DISCUSSION

In the current study, we examined the frequency of use of allied-health services among emerging and middle-aged adults with ASD from six Latin American countries, according to a sample of highly educated, computer literate parents almost all with insurance coverage. According to these parents, between 84.4%–95.9% were receiving zero hours of services per week. Moreover, for those having some services, 3.7%–12.9% receive between 1 and 10 h per week, and only 0%–2.7% 11 or more hours per week. Parents in our sample reported a very different experience than a prior parallel study of young people (Montiel-Nava et al., 2020). This is an unfortunate reality for adults with ASD in Latin America regarding the use of health services.

Although behavioral therapy was the most frequently used service in our sample, most participants were receiving none. Even considering that developed countries also report widely underserving adults with ASD (Dietz et al., 2020; Friedman et al., 2013; Płatos & Pisula, 2019; Shattuck et al., 2011; Vogan et al., 2017), the lack of services in Latin America is striking, bordering the total absence of services.

Families with an autistic adult member also confront health providers who are vastly unaware of ASD and its

TABLE 2 Clinical features

	Emerging adults (age 18–25) <i>n</i> = 201	Middle-aged adults (age 26–50) <i>n</i> = 94	Total <i>n</i> = 295	<i>x</i> ²	T
Autism diagnosis					
ASD % (n)	21.9 (44)	23.4 (22)	22.4 (66)	0.1	
Asperger % (n)	36.3 (73)	31.9 (30)	34.9 (103)	0.6	
Autism/autism spectrum % (n)	26.4 (53)	34 (32)	28.8 (85)	1.8	
PDD-NOS % (n)	10.9 (22)	7.4 (7)	9.8 (29)	0.9	
PDD % (n)	4.5 (9)	3.2 (3)	4.1 (12)	2.7	
Age of diagnosis mean $+$ SD	8.6 + 6.0	10.43 + 11.4	9.18 + 8.1		-1.8
Medical comorbid diagnoses					
Down syndrome % (n)	1.0 (2)	0 (0)	0.7 (2)	0.9	
Epilepsy % (n)	16.4 (33)	23.4 (22)	18.6 (55)	2.1	
Fragile—X % (n)	2 (4)	3.2 (3)	2.4 (7)	0.4	
Other % (n)	23.4 (47)	20.2 (19)	22.4 (66)	0.4	
Missing	56.2 (115)	53.2 (44)	55.9 (134)	_	
Count of medical comorbid disorders % (n)				1.9	
Only autism diagnosis/es	58.2 (117)	56.4 (53)	57.6 (170)		
One comorbid diagnosis	40.8 (82)	40.4 (38)	40.7 (120)		
Two comorbid diagnoses	1 (2)	3.2 (3)	1.7 (5)		
Severity index	1				
Low % (n)	18.4 (37)	23.4 (22)	20 (59)		
Moderate % (n)	40.8 (82)	38.3 (36)	40 (118)		
High % (n)	40.8 (82)	38.3 (36)	40 (118)		

^aVariable included missing cases.

manifestations during adulthood, leading to incorrect assumptions about their service needs (Shattuck et al., 2020; Zerbo et al., 2015). Such inaccurate clinical assumptions and inappropriate language might decrease their future autonomy (Kinnear et al., 2016; Nicolaidis et al., 2015). As adults with ASD are not offered enough services, they face long-term repercussions, including housing and employment opportunities (Butler & Gillis, 2011), relationships (Link & Phelan, 2001) denial of other types of services. Furthermore, it may lead to adverse health outcomes, limited self-efficacy, social isolation, which in turn makes it difficult accessing available services (Link & Phelan, 2001; Obeid et al., 2015; Martínez-Hidalgo et al., 2018; Muñoz et al., 2011; Stronach et al., 2019).

A unique result from this study is identifying the neurologists as the most frequently used service provider, even more regularly than psychiatrists or family doctors. The reported comorbidity with epilepsy in this sample could partially explain the high frequency of use of neurologists. According to the World Health Organization (2017), public expenditure on mental health and mental health specialists are scarce and unevenly distributed in Latin American and Caribbean countries (WHO, 2017), directly affecting access to services. Thus, a shortage of mental health specialists who attend to the adult

population with ASD might be somewhat responsible for reported services' paucity. On the other hand, in Brazil's health system neurologist rank as the second most available medical specialists in the country (Scheffer et al., 2018). Surprisingly, dietitians ranked second in the list of service providers. Without regional data to use as a baseline, we can only assume that non-healthy weight was the reason for this consultation. Reports of a significant association between obesity and ASD considered that autistic adults are more likely to be in non-healthy weight categories than non-autistic adults (i.e., underweight, overweight, or obese; Sedgewick et al., 2020; Zheng et al., 2017). In children, parents had reported the use of specialty diets to promote immune system function, increase quality and duration of sleep, and promote sociability among others (Trudeau et al., 2019). Thus, it might be possible that parents in our sample sought dietitians to develop food strategies or use dietary supplements to reduce symptoms or behaviors (e.g., specialty diets such as the ketogenic diet, low-carb diet, gluten free diet, and multivitamins). In general, lack of trained professionals, as it is the case of Latin American countries, is in the roots of delayed referral for services.

The medication in ASD is used for non-core symptoms (i.e., irritability, aggression), being the most used

TABLE 3 Usage of allied health services and use of medication

	Emerging adults (age: 18–25) <i>n</i> = 201	Middle-aged adults (age: 26–50) <i>n</i> = 94	Total <i>n</i> = 295	Т	x^2
Type of service (8 total)					
Speech therapy		_			0.1
0 h/wk % (n)	94 (189)	94.7 (89)	94.2 (278)		
1–10 h/wk % (n)	6 (12)	5.3 (5)	5.8 (17)		
11+ h/wk % (n)	0 (0)	0 (0)	0 (0)		
Occupational therapy					4.4
0 h/wk % (n)	92.5 (186)	86.2 (81)	90.5 (267)		
1–10 h/wk % (n)	7 (14)	11.7 (11)	8.5 (25)		
11+ h/wk % (n)	0.5 (1)	2.2 (2)	1.0 (3)		
Physical therapy					0.1
0 h/wk % (n)	95.5 (192)	95.7 (90)	95.6 (282)		
1–10 h/wk % (n)	4.5 (9)	4.3 (4)	4.4 (13)		
11 + h/wk % (n)	0 (0)	0 (0)	0 (0)		
Behavior therapy					1.4
0 h/wk % (n)	85.6 (172)	81.9 (77)	84.4 (249)		
1–10 h/wk % (n)	11.4 (23)	16 (15)	12.9 (38)		
11 + h/wk % (n)	3.0 (6)	2.2 (2)	2.7 (8)		
Cognitive rehabilitation					3.3
0 h/wk % (n)	91.5 (184)	89.4 (84)	90.8 (268)		
1–10 h/wk % (n)	6.5 (13)	10.6 (10)	7.8 (23)		
11 + h/wk % (n)	2.0 (4)	0 (0)	1.3 (4)		
Sensory integration therapy					1.4
0 h/wk % (n)	96.5 (194)	94.7 (89)	95.9 (283)		
1–10 h/wk % (n)	3 (6)	5.3 (5)	3.7 (11)		
11 + h/wk % (n)	0.5 (1)	0 (0)	0.3 (1)		
Social skills training					7.1*
0 h/wk % (n)	92 (185)	81.9 (77)	88.8 (262)		
1–10 h/wk % (n)	7 (14)	17 (16)	10.2 (30)		
11 + h/wk % (n)	1 (2)	1.1 (1)	1 (3)		0.5
Psychoanalytic services					
0 h/wk % (n)	90 (181)	92.6 (87)	88.8 (262)		
1–10 h/wk) % (n)	10 (20)	7.4 (7)	10.2 (30)		
11–20 h/wk) % (n)	0 (0)	0 (0)	0 (0)		
Medication received (Yes) % (n)	47.3 (95)	51.1 (48)	48.6 (143)	-0.57	
Total amount of allied health services $\%(n)^a$					6.6
No services	57.7 (116)	58.5 (55)	58 (171)		
One service	18.4 (37)	13.8 (13)	16.9 (50)		
Two services	9 (18)	9.6 (9)	9.2 (27)		
Three services	8.5 (17)	8.5 (8)	8.5 (25)		
Four services	2 (4)	2.1 (2)	2 (6)		
Five+ services	4.5 (9)	7.4 (7)	5.4 (16)		

*p < 0.05;

^aVariable included missing cases.

treatment for coexisting psychiatric conditions (Buck et al., 2014; Gotham et al., 2015; Fusar-Poli et al., 2019), and its use increases with age. In line with previous studies, almost half of the sample used medication. The difference in medication usage between children and adults in Latin America (37% vs. 49%) is not as alarming

	No medication ($n = 151$)	Medication ($n = 143$)	Total	<i>x</i> ²
Comorbid disorders % (n) ^a				4.3
Only an autism-related diagnosis	63.6(96)	51.7 (74)	57.8 (170)	
One comorbid diagnosis	35.1 (53)	46.2 (66)	40.5 (119)	
Two comorbid diagnoses	1.3 (2)	2.1 (3)	1.7 (5)	
Medication usage by epilepsy (most common comorbid DX) $\%$ (n) $^{\rm a}$	13.9 (21)	23.8 (34)	18.7 (55)	4.8*

**p* < 0.05;

^aVariable included missing cases.

TABLE 5 Medical and health providers

	Emerging adults (Age: 18–25) $n = 151$	Middle-aged adults (Age: 26–50) <i>n</i> = 143	Total <i>n</i> = 295	<i>x</i> ²
Medical & health providers $\%$ (n) ^a				
Audiologist ^a	7.7 (15)	3.3 (3)	6.3 (18)	2.0
CBT therapist ^a	15.7 (31)	14.1 (13)	15.2 (44)	0.1
Dietician ^a	37.2 (74)	32.5 (30)	35.7 (104)	0.6
Psychiatrist ^a	12.2 (24)	19.6 (18)	14.6 (42)	2.7
Neurologist ^a	46.2 (91)	40.4 (38)	44.3 (129)	0.9
School psychologist ^a	7.2 (14)	11 (12)	8.7 (25)	1.8

^aVariable included missing cases.

as the difference in allied health services (Montiel-Nava et al., 2020). The survey did not inquire about type of medication used, preventing further analysis. However, medication might have also been tailored for the medical comorbidities included in the "Other" category (i.e., ADHD, genetic, and congenital disorders). As families in Latin America frequently face a lack of services and trained professionals, medication might be seen as an alternative to help their members on the spectrum. Moreover, doctors could use medication as a palliative measure to help adults cope with the daily challenges associated with autistic symptoms and comorbid conditions (Fusar-Poli et al., 2019).

Current evidence, along with these new data, highlights the many systemic barriers experienced by adult autistic individuals and their families in low and middleincome countries, hindering their ability to access and benefit from health services (Durkin et al., 2015; Montiel-Nava et al., 2020; Paula et al., 2020). One of the main barriers refers to policy restrictions set by insurance companies (; Day et al., 2020; Friedman et al., 2013; Hansel, 2013; Lake et al., 2014; Raymaker et al., 2017; Shattuck et al., 2020; Vogan et al., 2017). Our sample was highly educated, with private insurance and access to the internet to complete the online survey, which might be related to the sampling method's nature. Despite these advantages over the included countries' general population, most participants were in the "Zero Hours of Services per week" category, highlighting the current lack of health services for adults with ASD in Latin America.

Enrollment in private insurances allows families with high or middle income to have another option on top of the public health system. However, our results show they are still not receiving services. Significant areas of concern in adulthood might remain unaddressed due to insufficient insurance coverage (Hansel, 2013; Friedman et al., 2013; Richards et al., 2012) or to the increasing costs of services (Friedman et al., 2013). As we described earlier in this paper, even with universal health care, families face overbooked facilities, long waitlists, scarcity of medicines, and a lack of trained professionals (Kohn, Saxena, Levav, & Saraceno, 2004; Paula et al., 2020; Saldivia et al., 2004)). In summary, having universal coverage is not a warranty for access to services in participant countries. Moreover, less affluent families might have no choice but to wait to be serviced at public institutions delaying possible diagnosis and treatment of ASD.

Suppose that the described lack of services is the reality of adults on the spectrum in Latin American countries with some advantaged socioeconomic background (private insurance, access to internet and computer, highly educated caregivers). What would be the actual use of services in the vast majority of the population? Following this analysis, we conclude that adults with ASD living in Latin America and disadvantaged socioeconomic backgrounds would confront additional barriers to services (lack of awareness, financial burden, transportation problems, stigma, etc.). Furthermore, they would be at an increased risk of receiving low service standards (Broder Fingert et al., 2019). As a result, their outcomes, including autonomy, health, quality of life, and caregivers' burden, might be impacted. Although ASD is typically diagnosed in childhood, it is a long-life neurodevelopmental disorder with different symptom expression depending on the developmental trajectory. Developing an agenda for practice, policymaking, and research about health disparities in ASD would identify further needs or modify existing policies to befit the expanding needs of the aging ASD adult (Bishop-Fitzpatrick & Kind, 2017). Legal and special education advocates could also assist this population by requesting transitioning services to late adolescents and emerging adults with ASD within educational settings. Both emerging and middle-aged adults were highly underserved; however, allocating these services and educating autistic individuals about their options at an early age could reduce this health inequality. As Bailey (2012) aptly expressed, the challenge lies in ensuring that legislation, resource allocation, and research studies reflect the reality of adults with ASD. This is the first study in Latin America attempting to show the barriers to accessing services for families with an adult autistic member.

Identifying health service usage among adults is of utmost relevance, given the prevailing nature of this condition. As adults with ASD experience different challenges than children, they will encounter the absence of individualized treatments, uncertainty about employment opportunities or vocational development, and challenges navigating the health system. Also, comorbid symptomatology might appear due to the difficulties mentioned above (Cheak-Zamora et al., 2018). In that way, any standard of care should consider the heterogeneous needs of services by age groups regarding type and number of services.

To our knowledge, this is the first study identifying service-use among adults with ASD in Latin America. Our Latin American sample demonstrated next to inexistent health care usage highlighting socioeconomic and health disparities in service provision for ASD symptoms and related conditions in the region. Even when diagnosed during childhood, ASD is a lifelong neurodevelopmental condition, and adults with ASD will likely continue to exhibit the triad of symptoms across the lifespan. The lack of services places adults with ASD in Latin America at a higher risk of worse symptom severity than autistic adults from regions with broader access to services.

Our study contains limitations. First, being an online survey, the sampling framework excluded those without internet access, hindering generalization of the results to all caregivers of adults on the spectrum in Latin America. Such a research design needed a sampling framework that, by definition, left out those without internet even before the study started. Internet access is far from universal in the participant countries, ranging from 23%(Chile) to 48.2% (Uruguay) at the time of the study (Internet Live Stats, 2016). This sampling bias is also evident in the proportion of caregivers with a college degree or higher (89.1) and the proportion using only private insurance to cover services for their children with ASD (71.4%). Second, despite an overall large sample size, there was a marked difference in the sample size among countries, with limited participants from the Dominican Republic and Venezuela. Third, since there is no published consensus concerning the number of hours adults should receive for ASD-related health services, we generated categories by the total hours of weekly service hours. Besides, data from use of service providers could not contribute to the calculation of hours of services. Fourth, by following the same format as the original Caregivers Need Survey (Daniels et al., 2017), this study did not inquire about each medication's name or classification reported by the participants. Such information could have allowed us to investigate further if the medication received is related to autism, a comorbid disorder to autism, or untreated medical conditions. Another limitation of the survey is that the data from the use of service providers could not contribute to the calculation of hours of services. The survey only asked caregivers to check from a list with different service providers the ones they have used, with no information about the number of hours received from each provider.

Conclusion

Since most research on ASD has focused on children's populations, the study of health services' uses and needs on adults with ASD is in its infancy. Our findings indicated that both emerging adults and middle-aged adults living in Latin America are highly underserved across all the examined health services, with between 86% and 95% receiving zero hours of services per week.

Adults with ASD living in Latin America are significantly at higher risk for low utilization of health services, which we speculate could lead to worse outcomes for their quality of life. Before this study, we knew next to nothing about adults on the spectrum living in Latin America. It is of utmost relevance to develop culturally sensitive best practices to address the adults' needs on the spectrum and their families living in low- and middleincome countries. By taking into account caregivers' perspectives, we have valuable information to guide subsequent research studies and match our goals with those needs of the autism community in Latin America, specifically those with less access to services.

ACKNOWLEDGMENT

The authors thank Cecilia Montenegro for her comments to this manuscript.

ORCID

Cecilia Montiel-Nava Dhttps://orcid.org/0000-0002-4293-5416

REFERENCES

- American Psychiatric Association (APA) (2000). Diagnostic and statistical manual of mental disorders (4th ed., Text Revision), Washington, DC: APA.
- American Psychiatric Association. (2013). *Diagnostic and statistical manual of mental disorders* (5th ed.). American Psychiatric Publishing.
- Bagaiolo, L. F., Mari, J. J., Bordini, D., Ribeiro, T. C., Martone, M. C. C., Caetano, S. C., Brunoni, D., Brentani, H., & Paula, C. S. (2017). Procedures and compliance of a video modeling applied behavior analysis intervention for Brazilian parents of children with autism spectrum disorders. *Autism*, 21(5), 603–610. https://doi.org/10.1177/1362361316677718
- Bailey, G. (2012). Emotional well-being for children with special educational needs and disabilities: A guide for practitioners. Sage.
- Baxter, A. J., Brugha, T. S., Erskine, H. E., Scheurer, R. W., Vos, T., & Scott, J. G. (2015). The epidemiology and global burden of autism spectrum disorders. *Psychological Medicine*, 45(3), 601–613.
- Bishop-Fitzpatrick, L., & Kind, A. J. H. (2017). A scoping review of health disparities in autism spectrum disorder. *Journal of Autism* and Developmental Disorders, 47(11), 3380–3391. https://doi.org/ 10.1007/s10803-017-3251-9
- Bordini, D., Paula, C. S., Cunha, G. R., Caetano, S. C., Bagaiolo, L. F., Ribeiro, T. C., Martone, M. C. C., Portolese, J., Moya, A. C., Brunoni, D., Bosa, C., Brentani, H., Cogo-Moreira, H., & de Jesus Mari, J. (2020). A randomised clinical pilot trial to test the effectiveness of parent training with video modelling to improve functioning and symptoms in children with autism spectrum disorders and intellectual disability. *Journal of Intellectual Disability Research*, 64(8), 629–643. https://doi.org/10. 1111/jir.12759
- Braden, B. B., Smith, C. J., Thompson, A., Glaspy, T. K., Wood, E., Vatsa, D., Abbott, A. E., McGee, S. C., & Baxter, L. C. (2017). Executive function and functional and structural brain differences in middle-age adults with autism spectrum disorder. *Autism Research*, 10(12), 1945–1959.
- Broder Fingert, S., Carter, A., Pierce, K., Stone, W. L., Wetherby, A., Scheldrick, C., Smith, C., Bacon, E., James, S. N., Ibañez, L., & Feinberg, E. (2019). Implementing systems-based innovations to improve access to early screening, diagnosis, and treatment services for children with autism spectrum disorder: An autism spectrum disorder pediatric, early detection, engagement, and services network study. *Autism*, 23(3), 653–664.
- Brugha, T. S., McManus, S., Bankart, J., Scott, F., Purdon, S., Smith, J., Bebbington, P., Jenkins, R., & Meltzer, H. (2011). Epidemiology of autism spectrum disorders in adults in the community in England. Archives of General Psychiatry, 68(5), 459–465. https://doi.org/10.1001/archgenpsychiatry.2011.38
- Buck, T. R., Viskochil, J., Farley, M., Coon, H., McMahon, W. M., Morgan, J., & Bilder, D. A. (2014). Psychiatric comorbidity and medication use in adults with autism spectrum disorder. *Journal of Autism and Developmental Disorders*, 44(12), 3063–3071.
- Butler, R. C., & Gillis, J. M. (2011). The impact of labels and behaviors on the stigmatization of adults with Asperger's disorder. *Journal of Autism and Developmental Disorders*, 41(6), 741–749.
- Centers for Disease Control and Prevention (CDC). (2011). Survey of pathways to diagnosis and services. Retrieved from https://www.cdc.gov/nchs/slaits/spds.htm.
- Cheak-Zamora, N. C., Teti, M., & Maurer-Batjer, A. (2018). Capturing experiences of youth with ASD via photo exploration: Challenges and resources becoming an adult. *Journal of Adolescent Research*, 33(1), 117–145.
- Cheak-Zamora, N. C., & Thullen, M. (2017). Disparities in quality and access to care for children with developmental disabilities and multiple health conditions. *Maternal and Child Health Journal*, 21(1), 36–44.
- Courchesne, E., Campbell, K., & Solso, S. (2011). Brain growth across the life span in autism: Age-specific changes in anatomical pathology. *Brain Research*, 1380, 138–145.

- Croen, L. A., Zerbo, O., Qian, Y., Massolo, M. L., Rich, S., Sidney, S., & Kripke, C. (2015). The health status of adults on the autism spectrum. *Autism*, 19(7), 814–823. https://doi.org/10.1177/ 1362361315577517.
- Daniels, A. M., Como, A., Hergüner, S., Kostadinova, K., Stosic, J., & Shih, A. (2017). Autism in Southeast Europe: A survey of caregivers of children with autism spectrum disorders. *Journal of Autism and Developmental Disorders*, 47(8), 2314–2325.
- Daniels, J. L., Forssen, U., Hultman, C. M., Cnattingius, S., Savitz, D. A., Feychting, M., & Sparen, P. (2008). Parental psychiatric disorders associated with autism spectrum disorders in the offspring. *Pediatrics*, 121(5), e1357–e1362.
- Day, T. C., McNaughton, K. A., Naples, A. J., & McPartland, J. C. (2020). Self-reported social impairments predict depressive disorder in adults with autism spectrum disorder. *Autism*, 24(2), 297– 306. https://doi.org/10.1177/1362361319857375
- Dietz, P. M., Rose, C. E., McArthur, D., & Maenner, M. (2020). National and state estimates of adults with autism spectrum disorder. *Journal of Autism and Developmental Disorders*, 50, 4258– 4266. https://doi.org/10.1007/s10803-020-04494-4
- Durkin, M. S., Elsabbagh, M., Barbaro, J., Gladstone, M., Happe, F., Hoekstra, R. A., Lee, L. C., Rattazzi, A., Stapel-Wax, J., Stone, W. L., & Tager-Flusberg, H. (2015). Autism screening and diagnosis in low resource settings: Challenges and opportunities to enhance research and services worldwide. *Autism Research*, 8(5), 473–476.
- Elias, R., & White, S. W. (2018). Autism goes to college: Understanding the needs of a student population on the rise. *Journal of Autism and Developmental Disorders*, 48(3), 732–746.
- Espadas, C., Ballester, P., Londoño, A. C., Almenara, S., Aguilar, V., Belda, C., Pérez, E., & Peiró, A. M. (2020). Multimorbidity and psychotropic polypharmacy among participants with autism spectrum disorder with intellectual disability. *Psychiatry Research*, 292, 113321.
- Friedman, N. D., Warfield, M. E., & Parish, S. L. (2013). Transition to adulthood for individuals with autism spectrum disorder: Current issues and future perspectives. *Neuropsychiatry*, 3(2), 181–192.
- Fusar-Poli, L., Brondino, N., Rocchetti, M., Petrosino, B., Arillotta, D., Damiani, S., Provenzani, U., Petrosino, C., Aguglia, E., & Politi, P. (2019). Prevalence and predictors of psychotropic medication use in adolescents and adults with autism spectrum disorder in Italy: A cross-sectional study. *Psychiatry Research*, 276, 203–209.
- Gotham, K., Marvin, A. R., Taylor, J. L., Warren, Z., Anderson, C. M., Law, P. A., Law, J. K., & Lipkin, P. H. (2015). Characterizing the daily life, needs, and priorities of adults with autism spectrum disorder from Interactive Autism Network data. *Autism*, 19(7), 794–804. https://doi.org/10.1177/1362361315583818.
- Griffiths, A. J., Giannantonio, C. M., Hurley-Hanson, A. E., & Cardinal, D. N. (2016). Autism in the workplace: Assessing the transition needs of young adults with autism spectrum disorder. *Journal of Business and Management*, 22(1), S22.
- Hansel, K. (2013). Rethinking insurance coverage of "experimental" applied behavioral analysis therapy and its usefulness in combating autism spectrum disorder. *Journal of Legal Medicine*, 34(2), 215–233.
- Hirvikoski, T., Mittendorfer-Rutz, E., Boman, M., Larsson, H., Lichtenstein, P., & Bölte, S. (2016). Premature mortality in autism spectrum disorder. *The British Journal of Psychiatry*, 208(3), 232–238.
- Internet Live Stats. (2016). Internet users by country. *Internet Live Stats*. https://www.internetlivestats.com/.
- Jokiranta-Olkoniemi, E., Gyllenberg, D., Sucksdorff, D., Suominen, A., Kronström, K., Chudal, R., & Sourander, A. (2020). Risk for premature mortality and intentional self-harm in autism spectrum disorders. *Journal of Autism and Developmental Disorders*, 1–11. https://doi.org/10.1007/s10803-020-04768-x.
- Kennedy-Hendricks, A., Epstein, A. J., Mandell, D. S., Candon, M. K., Marcus, S. C., Xie, M., & Barry, C. L. (2018). Effects of state

autism mandate age caps on health service use and spending among adolescents. *Journal of the American Academy of Child & Adolescent Psychiatry*, 57(2), 125–131.

- Kinnear, S. H., Link, B. G., Ballan, M. S., & Fischbach, R. L. (2016). Understanding the experience of stigma for parents of children with autism spectrum disorder and the role stigma plays in families' lives. *Journal of Autism and Developmental Disorders*, 46(3), 942–953.
- Kohn, R., Levav, I., de Almeida, J. M., Vicente, B., Andrade, L., Caraveo-Anduaga, J. J., Saxena, S., & Saraceno, B. (2005). Mental disorders in Latin America and the Caribbean: A public health priority. *Pan American Journal of Public Health*, 18(4–5), 229–240. https://doi.org/10.1590/s1020-49892005000900002
- Kohn, R., Saxena, S., Levav, I., & Saraceno, B. (2004). The treatment gap in mental health care. *Bulletin of the World Health Organization*, 82(1), 858–866.
- Lake, J. K., Perry, A., & Lunsky, Y. (2014). Mental health services for individuals with high functioning autism spectrum disorder. *Autism Research and Treatment*, 2014, 1–9.
- LeClerc, S., & Easley, D. (2015). Pharmacological therapies for autism spectrum disorder: A review. *Pharmacy and Therapeutics*, 40(6), 389–397.
- Link, B. G., & Phelan, J. C. (2001). Conceptualizing stigma. Annual Review of Sociology, 27(1), 363–385.
- Liptak, G. S., Kennedy, J. A., & Dosa, N. P. (2011). Social participation in a nationally representative sample of older youth and young adults with autism. *Journal of Developmental & Behavioral Pediatrics*, 32(4), 277–283.
- Magaña, S., & Smith, M. J. (2006). Psychological distress and wellbeing of Latina and non-Latina white mothers of youth and adults with an autism spectrum disorder: Cultural attitudes towards coresidence status. *American Journal of Orthopsychiatry*, 76(3), 346–357.
- Mandell, D. (2018). Dying before their time: Addressing premature mortality among autistic people. *Autism*, 22(3), 234–235.
- Mandell, D. S., Morales, K. H., Marcus, S. C., Stahmer, A. C., Doshi, J., & Polsky, D. E. (2008). Psychotropic medication use among Medicaid-enrolled children with autism spectrum disorders. *Pediatrics*, 121(3), e441–e448.
- Martínez-Hidalgo, M. N., Lorenzo-Sánchez, E., García, J. J. L., & Regadera, J. J. (2018). Social contact as a strategy for self-stigma reduction in young adults and adolescents with mental health problems. *Psychiatry Research*, 260, 443–450.
- Mason, D., Ingham, B., Urbanowicz, A., Michael, C., Birtles, H., Woodbury-Smith, M., Brown, T., James, I., Scarlett, C., Nicolaidis, C., & Parr, J. R. (2019). A systematic review of what barriers and facilitators prevent and enable physical healthcare services access for autistic adults. *Journal of Autism and Developmental Disorders*, 49(8), 3387–3400.
- Montiel-Nava, C., Chacín, J. A., & González-Ávila, Z. (2017). Age of diagnosis of autism spectrum disorder in Latino children: The case of Venezuelan children. *Autism*, 21(5), 573–580. https://doi.org/10. 1177/1362361317701267
- Montiel-Nava, C., Cukier, S., Garrido, G., Valdez, D., Paula, C. S., García, R., Rosoli, A., Irarrázaval, M., & Rattazzi, A. (2020). Service encounters across the lifespan in individuals with autism spectrum disorders: Results from a multisite study in Latin America. *Research in Autism Spectrum Disorders*, 79, 101670.
- Montiel-Nava, C., & Pena, J. A. (2008). Epidemiological findings of pervasive developmental disorders in a Venezuelan study. *Autism*, 12(2), 191–202. https://doi.org/10.1177/1362361307086663
- Muñoz, M., Sanz, M., Pérez-Santos, El., & Quiroga, M. A. (2011). Proposal of a socio-cognitive-behavioral structural equation model of internalized stigma in people with severe and persistent mental illness. *Psychiatry Research*, 186(2-3), 402–408. https://doi.org/10. 1016/j.psychres.2010.06.019.
- Murphy, C. M., Wilson, C. E., Robertson, D. M., Ecker, C., Daly, E. M., Hammond, N., Galanopoulos, A., Dud, I.,

Murphy, D. G., & McAlonan, G. M. (2016). Autism spectrum disorder in adults: Diagnosis, management, and health services development. *Neuropsychiatric Disease and Treatment*, *12*, 1669–1686. https://doi.org/10.2147/NDT.S65455

- Nicolaidis, C., Kripke, C. C., & Raymaker, D. (2014). Primary care for adults on the autism spectrum. *Medical Clinics*, 98(5), 1169–1191.
- Nicolaidis, C., Raymaker, D. M., Ashkenazy, E., McDonald, K. E., Dern, S., Baggs, A. E., Kapp, S. K., Weiner, M., & Boisclair, W. C. (2015). "Respect the way I need to communicate with you": Healthcare experiences of adults on the autism spectrum. *Autism*, 19(7), 824–831.
- Obeid, R., Daou, N., DeNigris, D., Shane-Simpson, C., Brooks, P. J., & Gillespie-Lynch, K. (2015). A cross-cultural comparison of knowledge and stigma associated with autism spectrum disorder among college students in Lebanon and the United States. *Journal of Autism and Developmental Disorders*, 45(11), 3520– 3536. https://doi.org/10.1007/s10803-015-2499-1
- Paula, C. S., Cukier, S., Cunha, G. R., Irarrázaval, M., Montiel-Nava, C., Garcia, R., Rosoli, A., Valdez, D., Bordini, D., Shih, A., Garrido, G., & Rattazzi, A. (2020). Challenges, priorities, barriers to care, and stigma in families of people with autism: Similarities and differences among six Latin American countries. *Autism*, 24(8), 2228–2242. https://doi.org/10.1177/ 1362361320940073
- Paula, C. S., Fombonne, E., Gadia, C., Tuchman, R., & Rosanoff, M. (2011). Autism in Brazil: Perspectives from science and society. *Revista da Associacao Medica Brasileira*, 57(1), 2–5. https://www. ncbi.nlm.nih.gov/pubmed/21390445
- Platos, M., & Pisula, E. (2019). Service use, unmet needs, and barriers to services among adolescents and young adults with autism spectrum disorder in Poland. *BMC Health Services Research*, 19 (1), 587.
- Raymaker, D. M., McDonald, K. E., Ashkenazy, E., Gerrity, M., Baggs, A. M., Kripke, C., Hourston, S., & Nicolaidis, C. (2017). Barriers to healthcare: Instrument development and comparison between autistic adults and adults with and without other disabilities. *Autism*, 21(8), 972–984.
- Richards, C., Oliver, C., Nelson, L., & Moss, J. (2012). Self-injurious behaviour in individuals with autism spectrum disorder and intellectual disability. *Journal of Intellectual Disability Research*, 56(5), 476–489. https://doi.org/10.1111/j.1365-2788.2012.01537.x.
- Rogers, S. J., & Vismara, L. A. (2008). Evidence-based comprehensive treatments for early autism. *Journal of Clinical Child & Adolescent Psychology*, 37(1), 8–38.
- Roux, A. M., Shattuck, P. T., Rast, J. E., Rava, J. A., Edwards, A. D., Wei, X., McCracken, M., & Yu, J. W. (2015). Characteristics of two-year college students on the autism spectrum and their support services experiences. *Autism Research and Treatment*, 2015, 1–10.
- Rubin, L. I., Geller, R. J., Nodvin, J., Marcus, M., Howett, M., & Merrick, J. (2011). Break the cycle of environmental health disparities in vulnerable children. *Reviews on Environmental Health*, 26(3), 135–137. https://www.ncbi.nlm.nih.gov/pubmed/ 22206188
- Saldivia, S., Vicente, B., Kohn, R., Rioseco, P., & Torres, S. (2004). Use of mental health services in Chile. *Psychiatric Services*, 55(1), 71–76.
- Scheffer, M. C., Cassenote, A. J. F., Guilloux, A. G. A., & Dal Poz, M. R. (2018). Internal migration of physicians who graduated in Brazil between 1980 and 2014. *Human Resources for Health*, 16 (1), 21. https://doi.org/10.1186/s12960-018-0286-8
- Sedgewick, F., Leppanen, J., & Tchanturia, K. (2020). Autistic adult outcomes on weight and body mass index: A large-scale online study. *Eating and Weight Disorders*, 25(3), 795–801. https://doi. org/10.1007/s40519-019-00695-8
- Shattuck, P. T., Garfield, T., Roux, A. M., Rast, J. E., Anderson, K., Hassrick, E. M., & Kuo, A. (2020). Services for adults with autism spectrum disorder: A systems perspective. *Current Psychiatry Reports*, 22(3), 13.

- Shattuck, P. T., Narendorf, S. C., Cooper, B., Sterzing, P. R., Wagner, M., & Taylor, J. L. (2012). Postsecondary education and employment among youth with an autism spectrum disorder. *Pediatrics*, 129(6), 1042–1049.
- Shattuck, P. T., Wagner, M., Narendorf, S., Sterzing, P., & Hensley, M. (2011). Post-high school service use among young adults with an autism spectrum disorder. *Archives of Pediatrics & Adolescent Medicine*, 165(2), 141–146.
- Stronach, S., Wiegand, S., & Mentz, E. (2019). Brief report: Autism knowledge and stigma in university and community samples. *Jour*nal of Autism and Developmental Disorders, 49(3), 1298–1302.
- Stuart-Hamilton, I., & Morgan, H. (2011). What happens to people with autism spectrum disorders in middle age and beyond? Report of a preliminary on-line study. *Advances in Mental Health and Intellectual Disabilities*, 5(2), 22–28.
- Trudeau, M. S., Madden, R. F., Parnell, J. A., Gibbard, W. B., & Shearer, J. (2019). Dietary and Supplement-Based Complementary and Alternative Medicine Use in Pediatric Autism Spectrum Disorder. *Nutrients*, 11(8), 1783. https://doi.org/10.3390/nu11081783.
- Vogan, V., Lake, J. K., Tint, A., Weiss, J. A., & Lunsky, Y. (2017). Tracking health care service use and the experiences of adults with autism spectrum disorder without intellectual disability: A longitudinal study of service rates, barriers and satisfaction. *Disability and Health Journal*, 10(2), 264–270.
- World Bank. (2017). World Bank Annual Report 2017. Washington, DC: World Bank. Retrieved from http://documents.worldbank. org/curated/en/143021506909711004/World-Bank-Annual-Report-2017.
- World Health Organization. (2005). Mental Health ATLAS 2005. Geneva: World Health Organization. Retrieved from https://www. who.int/mental_health/evidence/atlas/global_results.pdf?ua=1.
- World Health Organization. (2017). Mental Health ATLAS 2017. World Health Organization. Retrieved from https://www.who.int/ mental_health/evidence/atlas/profiles-2017/en/.
- Zerbo, O., Massolo, M. L., Qian, Y., & Croen, L. A. (2015). A study of physician knowledge and experience with autism in adults in a

large integrated healthcare system. Journal of Autism and Developmental Disorders, 45(12), 4002–4014.

- Zheng, Z., Zhang, L., Li, S., Zhao, F., Wang, Y., Huang, L., Huang, J., Zou, R., Qu, Y., & Mu, D. (2017). Association among obesity, overweight and autism spectrum disorder: A systematic review and meta-analysis. *Scientific Reports*, 7(1), 11697. https:// doi.org/10.1038/s41598-017-12003-4
- Zielinski, B. A., Prigge, M. B., Nielsen, J. A., Froehlich, A. L., Abildskov, T. J., Anderson, J. S., Fletcher, P. T., Zygmunt, K. M., Travers, B. G., Lange, N., & Alexander, A. L. (2014). Longitudinal changes in cortical thickness in autism and typical development. *Brain*, 137(6), 1799–1812.
- Zuckerman, K. E., Sinche, B., Cobian, M., Cervantes, M., Mejia, A., Becker, T., & Nicolaidis, C. (2014). Conceptualization of autism in the Latino community and its relationship with early diagnosis. *Journal of Developmental and Behavioral Pediatrics*, 35(8), 522– 532. https://doi.org/10.1097/DBP.000000000000091

SUPPORTING INFORMATION

Additional supporting information may be found online in the Supporting Information section at the end of this article.

How to cite this article: Torres, A., Lecusay, D., Valdez, D., Rosoli, A., Garrido, G., Cukier, S., Paula, C. S., Garcia, R., Rattazzi, A., & Montiel-Nava, C. (2021). Use of allied-health services and medication among adults with ASD in Latin America. *Autism Research*, 1–12. <u>https://doi.org/</u> <u>10.1002/aur.2583</u>